Author’s response to reviews

Title: Parent-child proximity and personality: Basic human values and moving distance

Authors:

Stefan Stieger (stefan.stieger@uni-konstanz.de)

David Lewetz (davidlewe@gmail.com)

Version: 1 Date: 02 May 2016

Author’s response to reviews:

See cover letter attached.